Author’s response to reviews

Title: Ever dispense of prescribed allergy medication in children growing up close to traffic: a registry-based birth cohort.

Authors:

Anna Lindgren (anna.lindgren@med.lu.se)
Emilie Stroh (emilie.stroh@med.lu.se)
Kristina Jakobsson (kristina.jakobsson@med.lu.se)

Version: 2
Date: 16 June 2015

Author’s response to reviews:

Dear Editors,

We have now added line numbers and page numbers to the manuscript.